Author's response to reviews

Title: Post-Traumatic Stress Disorder and its predictors in Emergency Medical Service personnel: A cross-sectional study from Karachi, Pakistan

Authors:
Salima Kerai (salima.kerai@aku.edu)
Uzma Khan (Uzma.khan@aku.edu)
Muhammad Islam (mislam1977@gmail.com)
Nargis Asad (nargis.asad@aku.edu)
Junaid Razzak (junaid.razzak@jhu.edu)
Omrana Pasha (Omrana.pasha@aku.edu)

Version: 2 Date: 12 Aug 2017

Author’s response to reviews:

August 12, 2017

To,
The Editor
BMC Emergency Medicine

Subject: Revision of manuscript EMMD-D-17-00014 entitled “Post-Traumatic Stress Disorder and its predictors in Emergency Medical Service personnel, Karachi, Pakistan”.

Dear Editor,

Thank you for accepting our manuscript for potential publication and giving us the opportunity carry out ressential revisions. Revisions have been made in line with the reviewers’ comments with regards to language editing and grammatical tense through out manuscript.

We have sought professional language editing services for language correction. We hope that, in its revised version, you will find the manuscript suitable for publication.

The revised version of manuscript has been read and approved by all authors.

Thank you